Author's response to reviews

Title: The PSA testing dilemma: GPs' reports of consultations with asymptomatic men. A qualitative study.

Authors:

Alison Clements (alison.clements@dphpc.ox.ac.uk)
Eila Watson (eila.watson@dphpc.ox.ac.uk)
Tanvi Rai (tanvi.rai@dphpc.ox.ac.uk)
Colleen Bukach (colleen.bukach@dphpc.ox.ac.uk)
Brian Shine (brian.shine@orh.nhs.uk)
Joan Austoker (joan.austoker@dphpc.ox.ac.uk)

Version: 3 Date: 14 June 2007

Author's response to reviews: see over
The PSA testing dilemma: GPs’ reports of consultations with asymptomatic men. A qualitative study.

Thank you for the instructions about the formatting changes. These have been completed as shown below. Box 1 and Table 1 files will be removed from the figure upload page at the point of submission.

Changes made:

Added UK to both addresses on page 1

All underlining removed from all section headings

Table placement text – removed from page 4

Box 1 title changed to Table 1 – page 16
Box 1 ref on page 3 changed to Table 1

Table 1 title changed to Table 2 – page 17
Table 1 ref changed to Table 2 – page 4

Superscript numbers changed to [ …] in all references throughout the manuscript

With best wishes

Alison Clements